Author’s response to reviews

Title: Splenic peliosis with spontaneous splenic rupture - a report of 2 cases.

Authors:

Daniel J. Lashbrook (dan.lashbrook@hotmail.com)
Roger W. James (Roger.James@ruh-bath.swest.nhs.uk)
Andrea J. Phillips (andrea.phillips@ruh-bath.swest.nhs.uk)
Anthony G. Holbrook (Anthony.Holbrook@ruh-bath.swest.nhs.uk)
Andrew C. Agombar (acagombar@yahoo.co.uk)

Version: 3 Date: 14 June 2006

Author’s response to reviews:

1) Correspondence author contact address changed on page 1
2) Type of laparotomy clarified in both case histories (pages 4 & 6 respectively)
3) Major change and re-referencing to paragraph "Haemorrhage associated with splenic peliosis...." (page 9) including reference suggested by reviewer.
4) Case history 1 shortened (removal of past medical history negatives and non-contributory routine blood results)